Author's response to reviews

Title: Attitudes towards terminal sedation: an empirical survey among experts in the field of medical ethics

Authors:

Alfred Simon (asimon1@gwdg.de)
Magdalene Kar (m.kar@arcor.de)
Jose Hinz (jhinz@zari.de)
Dietmar Beck (dbeck@zari.de)

Version: 4 Date: 12 April 2007

Author's response to reviews:

According to your email from April 4, we indicated only two authors as having contributed equally to the work.